DRAFT_1
EO-1 Weekly Status Week of March 24–March 30, 2011
Day of Year 083 - 089
Mission Day 3786 – 3792
Earth Observing One (EO-1) - General
Scheduled 106 science Data Collection Events (DCEs) this week.  
INSTRUMENTS

All instruments operated nominally this week 
· Performed instrument decontamination cycles

· Conducted Hyperion deicing on March 29 from 02:25z to 17:15z

· Conducted ALI outgassing on March 29 from 02:35z to 17:25z

· Performed Instrument calibration

· Conducted Solar Array Characterization on March 29 at12:32z

· Conducted Hyperion Solar Calibration on March 28 at 02:04z

Command and Data Handling (C&DH)

EO-1 has continued to experience problems playing back engineering data from the Solid-State Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.
On Thursday, March 24, there was an ESTO AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Steve Chien, Vuong Ly, Justin Rice, and Joe Young.

This teleconference focused on activities related to submitting a NASA Software of the Year Award for SensorWeb Tools.  Notes from this teleconference are as follows:

1. WEKA to WCPS translator is to be one of the elements in the SensorWeb Tools project.
2. There are three persons at JPL that are to contribute to the award submittal.  They are Steve Chien, Ashley Davies, and Danny Tran.

3. First input for award is to be submitted April 1.
4. Material is to be submitted to the GSFC Award Panel mid April.

5. Final award material is to be submitted late May/early June.

6. There are to be telecon’s with JPL on March 29 and April 6 to discuss their input to the submittal.
7. Dan Mandl is to work with Steve Chien the 4th week of April to finalize JPL’s input.

8. JPL has possible useful documentation from their successful 2005 Software award they will send to us.
On Tuesday, March 29, Dan Mandl had a telecon with Steve Chien and Ashley Davies of JPL to discuss the NASA Software of the Year Award submittal material regarding documentation of how the SensorWeb Tools has been utilized for monitoring volcanoes.   
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Setting up engineering passes with MGS and SG-1 to test ground stations after maintenance

· MGS certification supports scheduled for the week of April 4

· SG-1 tests are scheduled for the week of April 4

Mission Planning

· Continued testing of ASPEN to replace the legacy MOPSS mission planning software

· Running ASPEN in conjunction with some normal daily backups from MOPSS

· Running the backup UDAPs through SCP and CMS

· Working with JPL on Lunar calibrations changes that need to occur in ASPEN

· TDRS update made by JPL is being certified

· TDRS UDAP is matching MOPSS

· Updates to precision and base file id made by JPL are being checked

· All but 2 of the file IDs are working now

· Working with JPL to fix the last ones

· Determining the time shifts needed to adjust JPL lunar calibration model

· Continued functional testing of ASIST SCP to replace legacy CMS mission planning software

· Developer came in and made a change to SCP

· Preliminary tests show that the on screen output and the hex output are matching CMS

· Comparing Hobart ftf files created on SCP with Hobart ftf files created on CMS

· Preliminary tests show the correct xml and hex output

· Comparing TDRS ftf files created on SCP with TDRS ftf files created on CMS

· Preliminary tests show the correct xml and hex output

· Compared the ftf files from SCP to the ftf files from CMS, both matched

· Changed an RTS on SCP to match CMS

· Dumping all onboard RTS to verify the SCP matches the spacecraft

Flight Dynamics
· Verified the new update to SNAS, no changes were necessary

· Looking into output of elevation file to determine the reason that the script to check .ipr files are not picking out the elevations for most ground stations
MOC Move

· Finalized presentation for the MOC move plan to present to NASA

· Had meetings to determine the timeline on the MOC move

· Made small adjustments to the new MOC layout

· Determining the metrics used to verify functionality after the move
System Administration
· Continued to maintain EO-1 Data Processing during absence of Tech Engineer.

· Created 3 level 0 archive tapes and 4 dlt tapes for the science community.

· Corrected and reprocessed scene/image data for 4 downlinks.

· Archived dps user’s mail as a log of at/cron jobs related to processing of scenes.

· Continued working on the upcoming EO-1 MOC relocation.

· Collaborated with hardware personnel and FOT to create a relocation schedule.

· Updated the EO-1 file transfer spreadsheet that will be used as a baseline for system verification.

· Attended weekly MOC move meeting to review and discuss the MOC relocation presentation.

· Resolved disk space issue on the primary FEDS computer by archiving data from that partition to another computer to increase free space.

· Completed installation and testing of ClamAV antivirus software on another computer in support of POAM #26.

· Completed applying the March 2011 updates/patches on the 5 EO-1 CNE computers. The patches included Adobe Reader, Adobe Flash, Firefox, and Microsoft products.

· Completed applying the March 2011 updates/patches on the 3 EO-1 IONet computers. The patches included Adobe Reader, Adobe Flash, Firefox, and Microsoft products.

GROUND AND SPACE NETWORK

Station Downtimes

· SG-1 has been declared red until further notice.
· MGS red for planned maintenance (test passes being conducted).
Operational Discrepancies
· None
UPCOMING EVENTS

· Inclination maneuvers on hold until further notice

· Next lunar calibration to be in mid-April
Imagery Status

Scenes and Engineering Cals planned for week of March 24, 2011-March 30, 2011      106 
Total scenes and engineering calibrations planned for entire mission – approximately 59,272
Total Scenes:  ALI scenes in the Level 0 archive              54,581                (as of March 30, 2011)

                         Hyperion scenes in the Level 0 archive    54,324               
PUBLICATIONS AND PRESENTATIONS STATUS 
430 publications 

295 presentations

53 articles and press releases[image: image1.png]
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